

## REVIEW HISTORY

Manuscript Title: *Exploring Sustainable Tourism through Semio-Linguistic Analysis: The Case of Europe's Top Destinations*

Submitted: 05 July, 2024

Accepted: 10 October, 2024

Published: 31.10.2024

Corresponding Author: Renzo Mocini

Email: renzo.mocini@uniroma1.it

### **PEER REVIEW**

Reviewer 1: Irena Skendo

Reviewer 2: Luca Valleriani